Author's response to reviews

Title: Relationships between foot type and dynamic rearfoot frontal plane motion

Authors:

Vivienne H Chuter (Vivienne.Chuter@newcastle.edu.au)

Version: 7 Date: 10 June 2010

Author's response to reviews:
Author list on the submitted manuscript has been changed to match author information.
Table title now appears above the table
Manuscript has been checked for typographical errors.